Author's response to reviews

Title: Total Aortic Arch Replacement under Intermittent Pressure-augmented Retrograde Cerebral Perfusion

Authors:

Hiroshi Kubota (kub@ks.kyorin-u.ac.jp)
Kunihiko Tonari (ktonari@ks.kyorin-u.ac.jp)
Hidehito Endo (hendo@ks.kyorin-u.ac.jp)
Hiroshi Tsuchiya (tuchinococo@yahoo.co.jp)
Hideaki Yoshino (yoshino@ks.kyorin-u.ac.jp)
Kenichi Sudo (sudok@ks.kyorin-u.ac.jp)

Version: 6 Date: 1 November 2010

Author's response to reviews:


We would like to resubmit our manuscript formatting changed according to your message. Thank you for your consideration.

Sincerely, Hiroshi KUBOTA MD, Kyorin Univ. Tokyo, Japan